Author's response to reviews

Title: Cell type-specific anti-cancer properties of valproic acid: independent effects on HDAC activity and Erk1/2 phosphorylation

Authors:

Kamil Gotfryd (kamil@sund.ku.dk)
Galina Skladchikova (Galina.Skladchikova@dako.com)
Eugene E Lepekhin (elepekhin@email.com)
Vladimir Berezin (berezin@sund.ku.dk)
Elisabeth Bock (bock@sund.ku.dk)
Peter S Walmod (psw@sund.ku.dk)

Version: 2 Date: 1 June 2010

Author's response to reviews: see over
To *BMC Cancer*  

June 1, 2010  

Dear Editor,

I hereby submit a revised version of manuscript 1755359475345299 entitled:

**“Cell type-specific anti-cancer properties of valproic acid: independent effects on HDAC activity and Erk1/2 phosphorylation”**

By

Kamil Gotfryd, Galina Skladchikova, Eugene A. Lepekhin, Vladimir Berezin, Elisabeth Bock and Peter S. Walmod.

The manuscript has undergone an extensive revision according to the requests and suggestions provided by the two referees.

The revised manuscript is uploaded in two versions. In one of these all applied changes are highlighted.

Specific comments and points of criticism raised by the referees are answered in a point-by-point manner in a separate document.

I hope you will find the revised manuscript acceptable for publication in *BMC Cancer*.

Sincerely,

Peter S. Walmod